Author’s response to reviews

Title: Characterization of human mesothelin transcripts in ovarian and pancreatic cancer

Authors:

Zhanat E. Muminova (zhana@uab.edu)
Theresa V. Strong (tvstrong@uab.edu)
Dr Denise R. Shaw (dshaw@uab.edu)

Version: 3 Date: 10 May 2004

Dear Editor:

We have made the following format changes, as requested:
1. The word "Address" was removed from the affiliations list.
2. Superscript numbers have been used to link author names with the specific affiliation, in the format requested.
3. "PhD" has been removed from the title page.
4. The header has been removed from the manuscript pages.
5. Each figure has been cropped to minimized white space around edges.

We hope these changes are satisfactory, and thank you again for consideration of our work for publication.

Yours,
Denise R. Shaw
dshaw@uab.edu